Author’s response to reviews

Title: Malignancy during pregnancy in Japan: an exceptional opportunity for early diagnosis

Authors:
Masayuki Sekine (masa@med.niigata-u.ac.jp)
Yoshiyuki Kobayashi (og.koba@gmail.com)
Takayuki Enomoto (enomoto@med.niigata-u.ac.jp)
Tsutomu Tabata (tabatat@clin.med.mie-u.ac.jp)
Tamotsu Sudo (reikeih@wine.ocn.ne.jp)
Ryuichiro Nishimura (r_nishim@wk9.so-net.ne.jp)
Koji Matsuo (makouji@hotmail.com)
Brendan Grubbs (Brendan.Grubbs@med.usc.edu)
Tomoaki Ikeda (t-ikeda@clin.med.mie-u.ac.jp)

Version: 2 Date: 26 Aug 2017

Author’s response to reviews:

We wish to express our appreciation to the reviewers for their insightful comments.

<Reviewer #2>

Comment
The authors have done a very nice job responding to the comments and the paper is also much more clearly written. My only suggestion is that the reference to the American Cancer Society Department of Epidemiology and Statistics reports may not be correct. There is an Epidemiology Research Program, and they publish Cancer Statistics... I would check to make sure that it is correct before publication.

Response
We wish to thank the Reviewer #2 for this comment. In accordance with the Reviewer’s comment, we have changed the following text to the Background (p.5, lines 11-12).

(e.g. Cancer Statistics of American Cancer Society's Epidemiology Research Program, a population-based cohort study from the Cancer Registry and the Medical Birth Registry of
Norway, and an international collaborative setting of institutional registry in Belgium, the Netherlands and Czech Republic)

We wish to thank the Reviewers again for their valuable comments.